%~MoulinÃ© par MaN_auto v.0.32.4 2024-08-21 11:51:17
\documentclass[CRPHYS, Unicode, XML, screen, biblatex]{cedram}

\addbibresource{crphys20240308.bib}


\usepackage{multirow}
%\usepackage{ulem}
\usepackage{makecell}
%\setcellgapes{1pt}
\renewcommand\theadfont{\normalsize}


\newcommand{\bfk}{\mathbf{k}}
\newcommand{\bfu}{\mathbf{u}}
\newcommand{\bfv}{\mathbf{v}}
\newcommand{\bfe}{\mathbf{e}}
\newcommand{\bfx}{\mathbf{x}}
\newcommand{\bfc}{\mathbf{c}}
\newcommand{\bfn}{\mathbf{n}}
\newcommand{\rFr}{\mathrm{Fr}}
\newcommand{\rRe}{\mathrm{Re}}
\newcommand{\rRi}{\mathrm{Ri}}
\newcommand{\rEk}{\mathrm{Ek}}
\newcommand{\rRo}{\mathrm{Ro}}
\newcommand{\rSc}{\mathrm{Sc}}
%\newcommand*\odier[1]{\textcolor{blue}{#1}}
%\newcommand*\DEL[1]{\textcolor{green}{\sout{#1}}}


%\newcommand{\bbR}{\mathbb{R}}
%\newcommand{\bbC}{\mathbb{C}}
%\newcommand{\bbZ}{\mathbb{Z}}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

%\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\CDRsetmeta{articletype}{review}

%\makeatletter
%\def\TITREspecial{\relax}
%\def\cdr@specialtitle@english{Geophysical and astrophysical fluid dynamics in the Laboratory}
%\def\cdr@specialtitle@french{Dynamique des fluides gÃ©ophysiques et astrophysiques au laboratoire}
%\makeatother{}



\title{Internal gravity waves versus inertial waves in the laboratory}

\alttitle{Ondes internes de gravitÃ© versus ondes inertielles au laboratoire}

\author{\firstname{Sylvain} \lastname{Joubaud}\IsCorresp}
\address{Ens de Lyon, CNRS, Laboratoire de Physique, Lyon, France}
\email[S. Joubaud]{sylvain.joubaud@ens-lyon.fr}

\author{\firstname{Samuel} \lastname{Boury}}
\address{UniversitÃ© Paris-Saclay, CNRS, FAST, 91405 Orsay, France}
\email[S. Boury]{samuel.boury@universite-paris-saclay.fr}

\author{\firstname{Philippe} \lastname{Odier}}
\address[1]{Ens de Lyon, CNRS, Laboratoire de Physique, Lyon, France}
\email[P. Odier]{philippe.odier@ens-lyon.fr}

\thanks{The authors acknowledge support from the Simons Foundation through Grant No.~651475.\linebreak S. Joubaud acknowledge the support from Institut Universitaire de France.}
\CDRGrant{651475}


\keywords{Internal gravity waves, Inertial waves, Stratified fluid, Rotating fluid, Triadic Resonant Instability, wave attractors}

\altkeywords{Ondes internes de gravitÃ©, Ondes inertielles, Fluides stratifiÃ©s, Fluides en rotation, InstabilitÃ© par rÃ©sonance triadique, attracteurs d'ondes}


\begin{abstract}
Density-stratified and/or rotating fluids are very common in geophysical and astrophysical flows and enable the propagation of respectively internal gravity waves and inertial waves. Their peculiar dispersion relation has the same mathematical form for both classes of waves and can lead to unexpected outcomes through amplification, resonance or non-linearities. Even though their dispersion relation is very similar, internal gravity waves and inertial waves have different structural characteristics and arise from distinct physical mechanisms. Understanding the analogies and the differences in their behaviors is crucial for studying their respective roles. In this review, we will describe laboratory experiments that have studied either inertial waves in rotating homogeneous fluids or internal gravity waves in non-rotating density stratified fluids to highlight both the similarities and the differences between these two types of waves. We will focus on linear and non-linear phenomena occurring for three different configurations: wave beams in 2D and in 3D geometry, axisymmetric waves, as well as wave attractors, a specific feature for these waves. In particular, we will describe the influence of these various configurations on the Triadic Resonant Instability (TRI).
\end{abstract}


\begin{altabstract}
Les fluides stratifiÃ©s en densitÃ© et/ou tournants sont trÃ¨s courants dans les Ã©coulements gÃ©ophysiques et astrophysiques et permettent respectivement la propagation d'ondes internes de gravitÃ© et d'ondes inertielles. MathÃ©matiquement, la relation de dispersion particuliÃ¨re a la mÃªme forme pour les deux classes d'ondes et peut conduire Ã  des rÃ©sultats inattendus via amplification, rÃ©sonance ou non-linÃ©aritÃ©s. MÃªme si leur relation de dispersion est trÃ¨s similaire, les ondes internes de gravitÃ© et les ondes d'inertie ont des caractÃ©ristiques structurelles diffÃ©rentes et rÃ©sultent de mÃ©canismes physiques distincts. Comprendre les analogies et les diffÃ©rences dans leurs dynamiques est crucial pour Ã©tudier leurs rÃ´les respectifs. Dans cette revue, nous dÃ©crirons des expÃ©riences en laboratoire qui ont Ã©tudiÃ© soit les ondes d'inertie dans un fluide homogÃ¨ne en rotation, soit les ondes internes de gravitÃ© dans un fluide stratifiÃ© en densitÃ© non tournant, afin de mettre en Ã©vidence Ã  la fois les similitudes et les divergences entre ces deux types d'ondes. Nous nous concentrerons sur les phÃ©nomÃ¨nes linÃ©aires et non linÃ©aires se produisant pour trois configurations diffÃ©rentes : les faisceaux d'ondes en gÃ©omÃ©trie 2D et 3D, les ondes axisymÃ©triques, ainsi que les attracteurs d'ondes, spÃ©cificitÃ© de ces ondes. En particulier, nous dÃ©crirons l'influence de ces diffÃ©rentes configurations sur l'instabilitÃ© triadique rÃ©sonante (TRI).
\end{altabstract}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\dateposted{2024-11-07}
\begin{document}
\maketitle



\section{Introduction}

Internal gravity waves and inertial waves are two distinct classes of waves that commonly occur in fluid dynamics, respectively for density stratified fluids and rotating fluids~\cite{lighthill:78}. On the one hand, internal gravity waves correspond to disturbances in the density structure of the fluid, which leads to buoyancy forces acting as a restoring force. In a non-rotating density stratified fluid, the energy flux of pure monochromatic internal gravity waves propagates in a direction tilted by an angle $\beta$ with the horizontal, given by the dispersion relation~\cite{mowbray1967a,Sutherland:10}
\begin{equation}\label{disp_rel:Internal}
\omega = N\sin\beta,
\end{equation}
with $N$ the buoyancy frequency, which characterizes the stratification of the fluid.\newline On the other hand, pure inertial waves result from the action of the Coriolis force leading to an energy flux for monochromatic waves also inclined by an angle $\beta$ with the horizontal defined by the dispersion relation~\cite{Greenspan1968}
\begin{equation}\label{disp_rel:Inertial}
\omega = 2\Omega\cos\beta,
\end{equation}
with $\Omega$ the solid rotation rate of the fluid. As a comparison to the vertical density stratification, Coriolis force in rotating flows leads to a horizontal stratification in angular momentum.

In various geophysical and astrophysical phenomena, the dynamics is often based on a coupling between rotation and gravity. Such flows are therefore dominated by inertia-gravity waves, which is a combination of internal gravity waves and inertial waves. This type of waves share similar properties since their dispersion relation explicitly involves a coupling between the two characteristic frequencies $N$ and $f=2\Omega$
\begin{equation}\label{disp_rel:Internal-inertial}
\omega^2 = N^2\sin^2\beta+f^2\cos^2\beta\,.\end{equation}

Internal gravity waves and inertial waves are fundamental phenomena observed in various fluid systems, including the ocean, atmosphere, and astrophysical environments. In the oceans, these waves play a crucial role in mixing water masses, redistributing nutrients and heat, and influencing oceanic currents and circulation patterns~\cite{Garrett2003,Wunsch:ARFM:04,MacKinnon:climate:2017,DeLavergne2019,Pollmann2020,Oberwolfach2022}. Atmospheric internal waves often arise in stably stratified layers, such as within the troposphere or the bottom boundary layer, and contribute to turbulence, cloud formation, and the propagation of weather systems~\cite{Becker:climate:2003,Kim:Atm:2003,Camara2016,Holt:2022,Oberwolfach2022}. Gravity and inertial waves are also observed in astrophysical environments, such as stars, where they influence stellar dynamics, energy transport, and magnetic field generation~\cite{Kerswell:1998aa,Ogilvie:Astro:2004,Sidery2008,LeBars:ARFM:2015,Bouffard2021}.

The large scale of fluids in geophysical and astrophysical bodies induces for the waves a cascade of mechanical energy resulting from wave-wave, wave-vortex, wave-topography and/or wave-mean flow interactions. In particular, Triadic Resonant Instability (TRI) is an important type of wave-wave interaction that plays a major role in various geophysical contexts, and that has been extensively studied in the literature~\cite{Dauxois:ARFM:18}. TRI is characterized by the non-linear generation of two subharmonic secondary waves from a monochromatic primary wave, forming a triad of temporally and spatially resonant waves. We will come back throughout this article on various features of this instability. In general, these interactions result in an energy transfer from large scale to small scale structures so that waves can break and lead to turbulence and mixing or are dissipated through viscosity and diffusivity~\cite{Staquet:ARFM:02}.

Understanding these waves is therefore crucial for modeling and predicting the behavior of fluid systems across various scales. Interestingly, the dispersion relations for gravity waves~\eqref{disp_rel:Internal}, inertial waves~\eqref{disp_rel:Inertial}, and inertia-gravity waves~\eqref{disp_rel:Internal-inertial} share similar properties. In particular, the phase and group velocities are orthogonal for plane waves and no characteristic length is prescribed. This mathematical analogy between rotating and stratified fluids has been well-known for more than 75 years. In his review, Veronis~\cite{Veronis1970} describes this parallel and shows in different examples how results from one system can be analyzed and understood in terms of the behavior in the analogous system. More recently, the similarities among models for rotating fluids, stratified flows and even flows in magnetohydrodynamics have been used to understand in a unified manner the problem of weak turbulence of, respectively, inertial, gravity, and Alfv\`{e}n waves~\cite{Medvedev2010}.
Differences, however, still exist and have to be taken into account. The major one is the motion of the fluid particles in a wave. Indeed, for a plane wave, the wave is made of rectilinear oscillations for internal gravity waves while it is made of anticyclonic circular oscillations for inertial waves, implying a 3D motion in the latter case, contrary to the former (see in Maurer~\cite[Fig.~I.3]{Maurer:these:17}). Since the first experimental validation of the dispersion relation for internal gravity waves~\cite{Mowbray:JFM:67} and for inertial waves~\cite{Peacock:PoF:05}, they have been the object of many theoretical, numerical, and experimental studies in the last decades. In this review, we discuss recent experimental studies on either inertial waves or gravity waves in different setups using either non-rotating stratified fluids or rotating homogeneous (unstratified) fluids; a few cases of rotating-stratified configurations are also discussed. We focus on three main setups: wave beams, axisymmetric waves, and, finally, wave attractors. We will pay particular attention to studies involving Triadic Resonant Instability (TRI).



\subsection{Governing equations}

In this article, we consider a linearly stratified incompressible fluid subject to a global rotation at a rate $\Omega = f/2$ around the vertical axis, defined by the unit vector $\bfe_z$. We consider the case in which the background stratification is constant and parallel to the axis of rotation, i.e. the background density $\bar{\rho}$ is a linear function of $z$, characterized by the constant buoyancy frequency $N=\sqrt{-\frac{g}{\rho_0}\frac{\partial \bar{\rho}}{\partial z}}$ with $\rho_0$ the mean density. In the rotating frame of reference, the dynamics of such flow is described by the Navier--Stokes  and mass conversation equation under the Boussinesq approximation
\begin{align}
\rho_0\left(\frac{\partial \bfv}{\partial t}+\left(\bfv\cdot\nabla\right)\bfv\right)&=-\rho_0 f \bfe_z \times \bfv-\nabla p' - \rho' g\bfe_z+\rho_0\nu\Delta\bfv,\label{gene:NS}
\\
\frac{\partial \rho'}{\partial t}-\frac{\rho_0}{g}N^2\bfv\cdot\bfe_z+\bfv\cdot \nabla \rho'&=\kappa \Delta \rho',\label{gene:MC}
\\
\nabla\cdot \bfv&=0\label{gene:inc}
.
\end{align}
In these equations, $\bfv(\bfx,t)$ stands for the velocity field, $\nu$ the kinematic viscosity, $\kappa$ the diffusivity of the stratifying agent. The pressure field $p'$ is the pressure perturbation to the hydrostatic pressure incorporating the centrifugal acceleration. The density field $\rho'$ is the difference between the total density field $\rho$ and {the density background stratification $\rho'(\bfx,t)=\rho(\bfx,t)-\bar{\rho}(z)$}. As a result, the system forms a set of five equations with five unknown quantities: the velocity field $\bfv$, the pressure perturbation $p'$ and the density perturbation $\rho'$. In a given problem, ($\bfv$, $\rho'$, $p'$) are the solutions of equations~\eqref{gene:NS}-\eqref{gene:inc} completed with boundary conditions.

Introducing $U$ a characteristic velocity amplitude and $L$ a characteristic length, the previous equations exhibit various dimensionless numbers, which are important to characterize the flows. In what follows, we will consider waves and therefore, the characteristic length will be related in most cases to the wave length, $2\pi/|\bfk|$, with $\bfk$ the wave vector. Due to the anisotropy of rotating and/or stratified flows, it is important to consider separately the components that are perpendicular and parallel to the direction of the stratification/axis of rotation. We denote $k_\perp=2\pi/\lambda_\perp$ (respectively $k_z=2\pi/\lambda_z$) the perpendicular component (respectively parallel component) to $\bfe_z$. For each dimensionless number described below, two can be defined using the horizontal or the vertical wavelength, the subscript ${}_j$ standing for ${}_\perp$ and ${}_z$. Note that, in most stratified turbulence theories, the anisotropic limit $\lambda_z \ll \lambda_\perp$ is considered~\cite{Brethouwer:2007aa}. This limit is indeed relevant for geophysical flow. On the contrary, in experiments, $\lambda_\perp\sim\lambda_z$.
\begin{itemize}
\item The Froude number, $\rFr_j=\frac{U}{\lambda_j N}$, represents the ratio between the buoyancy time scale and the inertial time scale. The regime where internal gravity wave propagation takes place is the small Froude number regime.
\item The Rossby number, $\rRo_j=\frac{U}{2\Omega \lambda_j}$, compares the advection term to the Coriolis term. The regime where inertial wave propagation takes place is the small Rossby number regime.
\item The ratio $N/2\Omega$ compares the importance of stratification with rotation.
\item The Reynolds number, $\rRe_j=\frac{U\lambda_j}{\nu}$, characterizes the separation between laminar and turbulent flow.
\item The Schmidt number, $\rSc=\nu/\kappa$, compares the viscous effects with the diffusive ones. In the ocean, for salt water, $\rSc\approx 700$.
\item The Ekman number is defined as $\rEk=\frac{\nu}{2\Omega L^2 }=\frac{\rRo}{\rRe_j} \frac{\lambda_j^2}{L^2}$ with $L$ a characteristic global scale of the flow. It characterizes the importance of viscous forces compared with Coriolis forces. It is associated to phenomena emerging from boundary layers.
\item The Richardson number, $\rRi=\frac{g}{\rho_0}\frac{\vert\partial_z \rho'\vert}{(\partial_z U)^2}$, compares the potential energy due to buoyancy with {eddy kinetic energy}. Small Richardson numbers are associated to overturning and mixing~\cite{Ivey:ARFM:2008, Caulfield:ARFM:20}.
\end{itemize}
An interesting regime is the one for which ${\rRe \gg 1}$ while ${\rFr\ll 1}$, with low-frequency waves ${\omega/N \ll 1}$. This regime corresponds to stratified turbulence. For forcing frequencies $\omega/N$ comparable to the Froude number, the dynamics falls in the strongly stratified turbulence regime, relevant notably for the atmosphere. If $\omega/N$ is large compared to the Froude number, i.e. if ${\rFr \ll \omega/N \ll 1}$, the dynamics is dominated by weakly interacting non-linear internal gravity waves and corresponds to a weak wave turbulence regime. These regimes are the object of the review~\cite{Cortet:CRAS:2024}.

In the linear inviscid limit, this set of equations has wave solutions with the following dispersion relation
\begin{equation}\label{Disp_rel_global}
\omega^2 = N^2 \sin^2\beta+f^2\cos^2\beta=f^2 \frac{k_z^2}{k^2}+N^2\frac{k_\perp^2}{k^2},
\end{equation}
The angle $\beta$ is the angle between the vertical and the wave vector. This dispersion relation is illustrated in Figure~\ref{figBoury:polar} using ($\omega$, $\beta$) polar plane. The angle $\beta$ is associated to a single value $\omega$ while a value of $\omega$ is almost always associated to $4$ values of the angle $\beta$ corresponding to the $4$ quadrants of the oft-cited Saint Andrews cross. In the special case $f=N$, only a single value $\omega=N=f$ is allowed and valid for all angles. This schematic illustrates the symmetry between $N$ and $f$.

\begin{figure}[t]
\includegraphics[width=0.9\linewidth]{fig1.pdf}
\caption{Dispersion relation~\eqref{Disp_rel_global} plotted in the ($\omega$, $\beta$) polar plane, for dimensionless parameters $N$ and $f$: to each angle $\beta$ corresponds a unique frequency $\omega$ given by the blue curve (extracted from~\cite{BouryPhD2020}).}
\label{figBoury:polar}
\end{figure}

The dispersion relation~\eqref{Disp_rel_global} leads to different features. First of all, it limits the accessible values of the frequency $\omega$ to either $f<\omega<N$ or $N<\omega<f$. Outside these ranges, internal waves are evanescent. The low frequency domain can however be populated by other types of waves, such as Rossby waves which exist in the presence of a zonal mean flow. Another feature of these types of waves, the phase velocity $\bfc_\varphi$, aligned with the wave vector, and the group velocity $\bfc_g$ are orthogonal.
Importantly, as already mentionned in the introduction, the value of $\omega$ does not determine the wave length, which depends on many phenomena such as boundary effects or non-linear effects. This final remark implies that a particular attention must be paid to the choice of the wave maker to generate controlled gravity or inertial wave beams, as will be discussed in Section~\ref{exp_tech}.

An important feature in the comparison between internal gravity and inertial waves is the wave structure itself. To explain this difference, let us first consider a plane wave characterized by a single wave vector $\bfk_0$. The fluid motion for internal gravity waves is rectilinear, perpendicular to $\bfk_0$, and contained in the $(\bfk_0, \bfe_z)$ plane. In contrast, due to the Coriolis term, the inertial wave field is characterized by anticyclonic oscillations in the plane perpendicular to $\bfk_0$ and is therefore inherently three dimensional. In the following, for both inertial and internal gravity waves, we will call 2D wave any superposition of plane waves for which all the wave vectors are localized in a single plane containing $\bfe_z$. The velocity field will therefore be uniform in the direction normal to this plane. Note that, with this definition, an inertial wave will be considered as a 2D wave even if the three components of the velocity field are non zero. Such differences could lead to different behaviors as we will describe in the rest of this article.



Dissipation effects take place through viscosity or salt/temperature diffusion. In almost all experiments, this influence of viscosity has a negligible effect on the relation between $\omega$ and the direction of the wave vector. However, as the waves propagate, viscosity results in energy dissipation and attenuation of the wave. This dissipation has a characteristic time scale $1/(\nu k^2)$ and is therefore large for short wave lengths. It explains why viscosity has to be taken into account in laboratory experiments compared to geophysical and astrophysical flows. A self-similar solution describing the propagation of a viscous wave beam was found independently for rotating fluids~\cite{MooreSaffman1969} and for density-stratified fluids~\cite{Thomas:1972aa}. Near boundaries, such as the ocean floor or a container wall, viscosity plays a significant role in modifying gravity/inertial wave behavior. The dissipation near topographies clearly matters in the global energy budget~\cite{Nikurashin:JPO:10a}. Both inertial waves and gravity waves can be affected by the presence of a viscous boundary layers, which in turn can be modified due to the presence of gravity/inertial waves, leading to intricate patterns of motion and transport within these fluid systems~\cite{MandersMaas:2003,Passaggia:2014aa,Davis:JFM:19,Horne:JFM:19,Le-Dizes:2020aa}. For internal gravity waves, a strong mean flow will modify the wave field and may in turn be generated due to non-linearities of internal gravity waves in 3D~\cite{Bordes:PoF:12b,Dauxois:ARFM:18} or due to streaming in the viscous boundary layer as predicted theoretically~\cite{Renaud:2019aa} and observed experimentally~\cite{Horne:JFM:19}. The equivalent feature for rotating flows is the existence of zonal flows, a characteristic phenomenom of forced rotating flows. They are generated by different mechanisms such as non-linearities in the Ekman boundary layer~\cite{Calkins2010} or non-linear inertial waves~\cite{Tilgner:GAFD:07}. Such axisymmetric mean flows have been observed experimentally~\cite{Morize2010}.

Finally, an important phenomenon that needs to be considered in stratified fluids is internal-wave driven mixing~\cite{Onuki:2021aa}. When the wave amplitude is large enough, the shear associated with it can be characterized by a small Richardson number ($\rRi<0.25$), inducing overturns that can lead to mixing. This can be particularly true if TRI is present, contributing to energy transfers to smaller scales, resulting in a more efficient mixing. This mixing can be seen as a problem in experiments since the linear stratification cannot be conserved contrary to numerical simulations. However, mixing is critical for climate models~\cite{Whalen:2020aa}. In numerical models, mixing and associated turbulence are usually linked to subgrid scale internal wave processes. It is therefore crucial to understand the mixing caused by internal waves and to properly parameterise the mixing efficiency. To that extent, various experiments have been performed at ENS de Lyon to study the influence of TRI on mixing~\cite{Dossmann:EF:16,Dossmann:JGR:17}.



\subsection{Experimental techniques}\label{exp_tech}

Internal gravity waves in the laboratory require a stratified medium, which is generally achieved using salt as the stratifying agent, easier to manipulate than temperature. The filling of the tank is performed via the so called two-bucket method~\cite{Fortuin:JPS:60,Oster:CR:63}, which in its initial version produces a linearly stratified medium in a tank whose horizontal section is vertically uniform, but which can actually be generalized, using computer-controlled pumps, to create non-linear density profiles, or to produce a linear profile in a tank of vertically varying section~\cite{Hill:EF:02}. Stewart~et al.~\cite{Stewart:EF:21} have recently developed a derivation of this method, allowing to use temperature stratification.

As to rotating fluids, one ``simply'' needs to place a tank on top of a rotating table, which of course implies a number of technical challenges, including dealing with spin-up processes, non horizontal isobars, mechanically restraining the tank and all the measuring devices and allowing proper communication of power and data between the non-rotating and the rotating frame. The spin-up process, in particular, is rendered more complex when the fluid is also stratified~\cite{Maurer:these:17}, since the stratification prevents Ekman pumping motion, which is the major driving motion to attain solid body rotation during a spin-up.

Once the supporting medium is ready, a device is necessary to create the perturbation to this medium that will generate the wave. Almost 60 years ago, Mowbray and Rarity~\cite{mowbray1967a} were the first to produce internal waves in a laboratory experiment, using a sphere oscillating vertically in a linearly stratified fluid. Since then, experimentalists have been very creative in improving the design of the wave generators. Table~\ref{tab:gene} is an attempt to summarize and classify these various types of wave generators.

%\makecell[l]{}
In laboratory experiments, internal waves can be observed thanks to the following techniques:
\begin{itemize}
\item Dye lines, showing the motion of iso-density lines (or isopycnals), mainly qualitative;
\item Particle Image Velocimetry (PIV), based on following in time the motion of patterns of tracer particles, giving access to the velocity field;
\item Synthetic Schlieren, based on the apparent displacement of a pattern placed behind the experimental tank, linked to the change in refraction index of the fluid, associated with the density perturbation accompanying the wave propagation. It gives access to the density gradient field~\cite{Dalziel:EF:00} or to the whole density field~\cite{Oers:2023aa} (only possible for gravity waves);
\item Planar Laser Induced Fluorescence (PLIF), using a laser-excited fluorescent dye as tracer for density, allowing to measure directly the density field. It can be coupled to PIV, using the same laser plane, to get access to density-velocity correlations and therefore density fluxes, important for studies of mixing~\cite{Dossmann:EF:16} (only possible for gravity waves).
\end{itemize}


\begin{table}[!htp]
\caption{List of experimental gravity and/or inertial wave generators, organised by type of wave form generation.}
\centering
%\makegapedcells
\begin{tabular}{|l|l|l|l|l|}
\hline & \thead{\textbf{Category}} & \thead{\textbf{Description}} & \thead{\textbf{Type of wave}} & \thead{\textbf{Example} \\ \textbf{reference}} \\
%\ & & & & \thead{\textbf{}}\\
\hline
\multirow{2}{*}{I} & \multirow{2}{*}{Whole tank oscillation} & Up and down motion & Global mode or &~\cite{Benielli:DAO:96} \\
\cline{3-3}\cline{5-5}~&~& Libration & self-similar wave beam &~\cite{Noir:PEPI:12} \\ \hline

\multirow{2}{*}{II} & \multirow{2}{*}{Global boundary motion} & Oscillating rigid wall & \multirow{2}{*}{Guided wave} &~\cite{Davis:JFM:19,Savaro:PRF:20} \\ \cline{3-3}\cline{5-5}~&~& Oscillating rigid lid &~&~\cite{Pacary:PRF:23} \\ \hline

\multirow{7}{*}{III} & \multirow{7}{*}{Oscillating object} & Oscillating cylinder & \multirow{7}{*}{Self-similar wave beam} &~\cite{Mowbray:JFM:67,Dauxois:PoF:04} 
\\ 
\cline{3-3}\cline{5-5}~&~& Oscillating sphere &~&~\cite{Voisin:JFM11} \\ \cline{3-3}\cline{5-5}~&~& Oscillating torus &~&~\cite{Duran:PRE:13} \\ \cline{3-3}\cline{5-5}~&~& Multiple &~&~\cite{Monsalve:PRL:20} \\
%\cline{3-3}
\cline{5-5}~&~& oscillating objects &~&~\\
\cline{3-3}\cline{5-5}~&~& Tidal motion on &~&~\\
%\cline{3-3}
\cline{5-5}~&~& a fixed obstacle &~&~\cite{Gostiaux:PoF:07}\\ \hline
IV & Towed object & Towed object & Wave beam &~\cite{Scase:JFM:06} \\ \hline

\multirow{5}{*}{V} & \multirow{5}{*}{Oscillating plates} & Linear device & \multirow{5}{*}{Versatile wave form} &~\cite{Gostiaux:EF:07} \\
\cline{3-3}\cline{5-5}~&~& Independent &~&~\cite{Dobra:EF:19,Brunet:PRF:19,Husseini:PRF:20} \\
\cline{5-5}~&~& oscillating plates &~&\\
\cline{3-3}\cline{5-5}~&~& Axisymmetric &~&\\
\cline{5-5}~&~& oscillating plates &~&~\cite{Maurer:EF:17}\\ \hline

VI & Local boundary & \multirow{2}{*}{Deformable tank} & \multirow{2}{*}{Specific purpose} & \\
   &  deformation & & &~\cite{Malkus:GAFD:89,Favier:EAS:19}\\
\hline

\multirow{2}{*}{VII} & \multirow{2}{*}{Liquid interface motion} & Turbulent underlayer & \multirow{2}{*}{Uncontrolled} &~\cite{Munroe:EFM:08} \\ \cline{3-3}\cline{5-5}~&~& Convective underlayer &~&~\cite{Dorel:PRF:23}\\ \hline
\end{tabular}
\label{tab:gene}
%\vspace*{-10pt}
\end{table}
%}



The two most common techniques, PIV and Schlieren, allow quite easily for an extraction of wave frequency content and spatial structure, using respectively Fourier transform (or time frequency analysis~\cite{Flandrin:matlab:99} when time-dependence is relevant) and Hilbert transform~\cite{Mercier:PoF:08}.
\linebreak Many experiments require to estimate or to measure quantitatively the amplitude of the primary wave. One cannot rely on the amplitude of the wave generator motion, since there is an efficiency factor that comes into play and that depends strongly on frequency, wave form and experimental conditions. For the Schlieren technique, such measurement requires integration of the mass conservation equation, which in general cases with multiple waves is not a simple task. In the same way, for PIV measurements, a direct estimate of the kinetic energy of the whole velocity field can easily be computed. However, quantifying the amplitude of a specific wave among other waves in interaction can be difficult.



Among the various methods used to extract a wave amplitude, Mathur~et al.~\cite{Mathur:PRL:10} studied transmission and reflection of internal wave beams across a transmission region and took a Fourier transform of the reflected and transmitted wave fields along appropriately chosen transects. Maurer~\cite{Maurer:these:17} measured wave amplitudes by looking at the maximum of the velocity over a given spatial area, and Supekar~et al.~\cite{Supekar:JFM:19} utilized the distribution of maxima of amplitudes for a velocity field in a widespread two-dimensional beam. Boury et al.~\cite{Boury:PRF:19} first performed a fit of the wave form to the expected spatial profile as a function of time, thus extracting a wave amplitude as a function of time, before averaging in time the obtained amplitude, choosing a time window where the process was as stationary as possible. When the amplitude of two different waves of same frequency must be estimated (in the case of wave reflection for example), a method based on the signal processing technique called Variable Mode Decomposition has been adapted successfully~\cite{Horne:EF:21}. In the case where time stationarity is not achieved in the observed experiment, Boury~et al.~\cite{Boury:JFM:24} developed a method based on using Hilbert transform times series and root-mean-square averaging to obtain a time dependent amplitude of the waves. Finally, in the case where synthetic Schlieren is used to observe the waves, Grayson~et al.~\cite{Grayson:JFM:22}, based on previous developments~\cite{Scase:JFM:06,Hazewinkel:JFM:10}, used integrated components of the density gradient field, to which they apply a root-mean-square and spatial average.



\section{Internal Wave Beams in a Square Geometry and its Instability}

During the years 2010-2015, a series of studies~\cite{Bordes:PoF:12b,Bourget:JFM:13,Bourget:JFM:14} was undertaken by a group at ENS de Lyon, jointly with a group at UniversitÃ© d'Orsay (now UniversitÃ© Paris-Saclay), where the generation of plane beams of internal gravity waves (in Lyon) and inertial waves (in Orsay) was obtained using the same wave generator (linear oscillating plates, type V of Table~\ref{tab:gene}), allowing for interesting comparisons between both types of generated waves, and in particular regarding the Triadic Resonant Instability of the corresponding wave beams.



\subsection{Generation of Gravity Versus Inertial Waves}

In both cases, the generation of the waves was very similar, using the wave generator described in~\cite{Gostiaux:EF:07}, placed in a tank, either non-rotating but linearly salt stratified (in Lyon) or non stratified but lying on top of a platform rotating at rotation rate $\Omega$ (in Orsay). It was actually the same device, which travelled from Lyon to Orsay. In both cases, the generator configuration was set to excite a plane wave beam, with $3$ to $4$ wave lengths of about $7$ cm each along the whole height of the generator, with a tapering amplitude at each end to avoid spurious emission of internal waves.

However, a main difference lies in the generation process, since the fluid motion in the case of internal gravity waves is a simple oscillating translation in the direction of the group velocity, whereas fluid particles describe anticyclonic circular translation in the case of inertial waves. As a consequence, for inertial waves, the oscillating plates of the wave maker only force the longitudinal component of their circular motion, while the lateral component is let to freely adjust according to the spatial structure of the wave solution. This makes the generator less efficient in producing inertial waves, compared to gravity waves. A detailed description of the added difficulty when producing inertial waves with this generator, with an efficiency depending strongly on generator amplitude and wave length can be found in the appendix of~\cite[Chapter~3]{Bordes:these:12}.

In order to make up for that difficulty, the generator angle was adapted to the oscillating frequency $\omega$, so that the plate motion would actually be in the exact direction of the longitudinal component of the wave motion. Note that this solution is not applicable to the case of gravity waves, since reclining the generator, with a change of inclination for each frequency studied, would be complicated and would also rapidly destroy the stratification.

In the case of inertial waves, it turned out that in order to vary the wave angle, set by the ratio $\omega/f=\omega/2\Omega$, it was easier to change the global rotation of the table, rather than changing the generator frequency. For gravity waves, of course, changing the stratification was not an option and it is the wave frequency $\omega$ that was varied.

\begin{figure}[!ht]
\includegraphics[width=1\linewidth]{fig2.pdf}
\caption{Examples of frequency spectra showing TRI. (Left) Frequency spectrum for internal gravity waves ($N\neq0, f=0$), from~\cite{Bourget:JFM:13}. (Right) Frequency spectrum for inertial waves ($N=0, f\neq0$), from~\cite{Bordes:PoF:12b}. In the latter case, a narrow peak corresponding to the rotation frequency of the table (thus at $\omega/f=0.5$) is visible, linked to mechanical noise of the table.}
\label{fig:bordesbourget}
\end{figure}



One last point of comparison: in both cases, the generator was parallelepippedic. While for gravity waves, which have a perfect 2D behaviour, this is not a problem, it becomes more tricky with inertial waves, which have inherent 3D motion and the tank shape is not really adapted, with lateral walls restraining the transverse motion. For this reason, while the tank used for gravity waves was narrow ($2$ wave lengths), in the case of inertial waves, the tank was much wider, with a width of about 10 wave lengths.

As we will show later, in spite of these differences, the production of the waves and the characteristics of the instability via resonant interaction are very similar.




\subsection{Instability of the Wave Beams}\label{subsection:TRI}


Already in $1967$, Davis and Acrivos~\cite{Davis:JFM:67} showed experimentally and theoretically that, due to the non-linear terms of the wave equations, internal waves are unstable to infinitesimal perturbations, which grow to form temporally and spatially resonant triads between the primary wave (indicated in what follows by index $0$), and two secondary waves (indicated by indices $1$ and $2$). These resonant conditions write respectively
\begin{align}
\omega_{0} & = \omega_{1}+\omega_{2},\\
\bfk_{0} & = \bfk_{1}+\bfk_{2}.
\end{align}
This instability is now referred to as \emph{Triadic Resonant Instability} (TRI). The analytical derivation for the plane wave solutions of the non-linear wave equations can be found in Dauxois~et al.~\cite{Dauxois:ARFM:18}. For a given primary wave vector of components $(\ell_{0},m_{0})$, combining the dispersion relation of the waves with the temporal resonance condition (and eliminating one of the two secondary wave vectors using the spatial resonance condition), one obtains the following equation for the components $(\ell_{1},m_{1})$ of the remaining secondary wave vector
\begin{equation}\label{equfinale}
\frac{|\ell_0|}{\sqrt{\ell_0^2+m_0^2}} = \frac{|\ell_1|}{\sqrt{\ell_1^2+m_1^2}} + \frac{|\ell_0 - \ell_1|}{\sqrt{(\ell_0-\ell_1)^2+(m_0-m_1)^2}}\,.
\end{equation}
\label{equation_k1m1}
The set of solutions of this equation can be represented by a resonance locus (\cite[Figure~2.a]{Dauxois:ARFM:18}) defined as the possible locations in wave vector space of the tip of vector $\bfk_{1}$. We recall the typical shape of this locus, with examples of the vector triads $\bfk_{0}$, $\bfk_{1}$, and $\bfk_{2}$ in the insets of Figure~\ref{fig:sigmaK}$\MK$(left).

The TRI instability was observed in several laboratory experiments~\cite{McEwan:JFM:75,Benielli:JFM:98,Joubaud:PoF:12} with increasing details thanks to progresses in fluid motion imaging, using in particular PIV and synthetic Schlieren in the last decade. This instability was shown to play a role in oceanic energy transport and vertical mixing~\cite{Mackinnon:JPO:13c,Hu:JPO:23}.



Here again, the joined study performed by the Lyon and Orsay groups using the same wave generator provides an interesting view on the TRI instability in the respective case of internal gravity waves and inertial waves. In both cases, with a large enough amplitude -- typical wave velocity of the order of 2 to 3 mm.s$^{-1}$, or in other words dimensionless quantity $\psi_{0}/\nu$ (equivalent to a typical Reynolds number of the flow), of the order of a few tens, where $\psi_{0}$ is the stream function amplitude of the wave -- the instability starts after 20 to 40 generator periods.

Figure~\ref{fig:bordesbourget} shows the frequency spectra of the wave system after the onset of instability, on the left for the gravity waves and on the right for the inertial waves. Three main peaks are observed, corresponding to the primary and two secondary waves. In both cases, the temporal resonance condition is verified. However, the secondary peaks are much broader for the inertial wave case than for gravity waves. {A possible explanation for this observation is given in Section~\ref{sec:frustrum}}.




One of the interesting features of TRI, in the oceanic mixing context, is that it allows for a transfer of energy to smaller scales, without the necessity of a turbulent energy cascade. That happens in particular when both secondary waves have larger wave vectors, compared to the primary wave: $\kappa_{1,2}>\kappa_{0}$, where $\kappa$ is the modulus of the wave vector. This is the case, in general, when the solutions belong to the external branches of the resonance locus (see example in top inset of Figure~\ref{fig:sigmaK}$\MK$(left)). In contrast, when the solutions are on the central part of the locus, the transfer goes both directions, with $\kappa_{1}>\kappa_{0}>\kappa_{2}$, or vice-versa (see example in bottom inset of Figure~\ref{fig:sigmaK}$\MK$(left)).

\begin{figure}[!ht]
\includegraphics[width=1\linewidth]{fig3.pdf}
\caption{Normalized wave numbers $\kappa_{1,2}/\kappa_0$ of the secondary waves, as a function of the primary wave frequency $\omega_0$. (Left) Gravity waves, adapted from\cite{Bourget:JFM:13}. (Right) Inertial waves~\cite{Bordes:PoF:12b}. The lines represent theoretical predictions, computed for the wave dimensionless amplitude $\psi_{0}/\nu=28$ (left) and $\psi_{0}/\nu=37$ (right). The insets on the left plot show two typical examples of the resonance locus. As illustrated in this inset, the solid lines on the left plot (upward going dotted lines on right plot) represent the case where the secondary wave vectors are on the external branches of the resonance locus, while the dashed lines on the left plot (mostly downward going dotted lines on the right plot) represent the case where the secondary wave vectors are on the central part of the locus. Between these two types of solutions, the one that has the largest growth rate is represented by a thicker line. Filled circles and squares with error bars correspond to experimental measurements. The two solid thin lines on the right plot show the allowed range around the dotted curves, determined by considering an uncertainty of $\pm 50\%$ on the wave amplitude.}
\label{fig:sigmaK}
\end{figure}


Both studies mentioned earlier delved into this issue. For a given wave amplitude and for each normalised frequency $\omega/N$ of the primary wave, the relative magnitudes $\kappa_{1,2}/\kappa_{0}$ of the secondary wave vectors that correspond to the largest growth rate of the instability were computed, both in the case of the external and central branches of the resonance locus. The result is presented in Figure~\ref{fig:sigmaK}, on the left plot for internal gravity waves~\cite{Bourget:JFM:13} and on the right plot for inertial waves~\cite{Bordes:PoF:12b}. In addition, a thickening of the lines highlights the case (between central and external branches) with highest growth rate. There is a transition from the central to the external branch, at a frequency around $\omega/N=0.77$ for gravity waves and $\omega/f=0.79$ for inertial waves. Experimental points are also shown, with colored symbols. Interestingly, in both cases, even below the transition frequencies mentioned above, the experimental solutions seem to stay on the external branch, with, mostly, an energy transfer towards smaller scales, except at the lowest frequencies where TRI was observed in the case of inertial waves. Note that for gravity waves, were only one experimental point is shown, the magnitude of the wave vectors is slightly larger than predicted.



In a laboratory experiment of typical sub-meter size, the viscosity of water is such that small scale waves are damped quickly and secondary waves with wave vector magnitude above a few times the primary wave vector magnitude are never the most unstable solution (see~\cite[Figure~2.c]{Dauxois:ARFM:18}) and are therefore never observed. However, the opposite trend can be observed in Figure~\ref{fig:sigmaK}, where wave vectors longer than expected were measured and the solution on the central branch of the resonance locus, where wave vectors are shorter, was never observed.

This could be explained by a finite-size effect. This effect was described in Bourget~et al.~\cite{Bourget:JFM:14} and relates to the fact that in finite-size beams with a width equal to only a few wave lengths, the secondary waves, which have a group velocity with a different direction than the primary wave, leave the interaction region, located within the primary wave beam width, before the instability has time to grow. This effect is of course enhanced when the waves have a large group velocity, which is the case of waves of small wave vectors, since group velocity increases when wave vector amplitude decreases. For this reason, one can understand that the solutions on the central branches of the resonance locus, which, for infinitely wide beams, have a larger growth rate below the transition frequency, are not observed in the experiments, since the actual growth rate, taking into account the finite size effect, is reduced (see equation 3.7 in~\cite{Bourget:JFM:14}).



\subsection{Variability of the Triadic Resonant Instability}

A recent study by Grayson et al.~\cite{Grayson:JFM:22} focused on the long-term fate of TRI in the case of narrow beams of internal gravity waves. The authors used a large scale tank ($11$ m wide) to prevent interactions between the beam and its reflections and they took advantage of the very versatile ``magic carpet'' wave generator built in Cambridge (type V of Table~\ref{tab:gene},~\cite{Dobra:EF:19}). By doing a refined analysis of the characteristics of the secondary waves of the TRI, they showed that there exists an important variability in amplitude, frequency and wave vector of these secondary waves. This variability takes place both in time and space.

This is illustrated in Figure~\ref{fig:Greyson}, showing the power spectral density at the function of time and frequency (left plot) and as a function of the secondary wave vector horizontal components and of the height in the tank (right plots). On the left plot, one can observe that while the primary wave frequency maintains a constant value at the forcing frequency $\Omega_{0}/N$=0.62, both secondary frequencies fluctuate in time, while conserving the resonance property $\omega_{1}+\omega_{2}=\omega_{0}$. Note that frequencies around $0.0$, $0.8$, and $1.0$ come from higher order interactions. In order to better understand the underlying modal structure of the observed beams, Grayson et al. decomposed the flow field using Dynamic Mode Decomposition (DMD, white line superimposed in the time-frequency plot), confirming the observed variability. A similar variation in frequency has been witnessed by both Bourget~et al.~\cite{Bourget:JFM:13} and Brouzet~et al.~\cite{Brouzet:EPL:16} in their experimental studies. One can also observe at certain times (in particular between 100 and 200 $T_{0}$, as well as between 200 and 300 $T_{0}$), other pairs of secondary beams appearing and then merging with the continuous mode, in the form of convergent ``wisps'' on the $\omega_{1}/N$ and $\omega_{2}/N$ bands. The authors have checked that in such cases, the triadic resonance in frequency persists across all the spectrum.

\begin{figure}[th]
\includegraphics[width=0.99\linewidth]{fig4.png}
\caption{ (Left) Time-frequency spectra (log scale of normalised spectral density) for a long term experiment. The dominant frequencies obtained from the DMD frequency decomposition are overlaid in white. (Right) For the same experiment, distribution of energy in the $\ell/\ell_0$ space and over the non-dimensional height in the domain $z/H$. (top) $t/T_0 = 83$, (bottom) $t/T_0 = 481$. Both the surface plot colour and the height of the peaks show the power spectral density. The left (right) ``bump'' corresponds to $\ell_2$ ($\ell_1$). The background plot shows the power spectral density at the same instant in time in the Fourier plane of horizontal wave number component and frequency. Adapted from~\cite{Grayson:JFM:22}.}\label{fig:Greyson}
\vspace*{-5pt}
\end{figure}


The right plot shows, at two different times, the distribution of energy along the horizontal component of the wave vector and across the height of the tank. One can observe that at the earlier time (top figure), the TRI is mostly localised in the upper part of the tank, with the $\ell_{2}$ component at a value around $-2 \ell_{0}$ and the $\ell_{1}$ component at a value around $0.8 \ell_{0}$. Note that the two secondary waves can be identified on this plot thanks to the background plot, showing the distribution of energy in the ($\ell,\omega$) plane, the wave with largest frequency being identified with index 2 ($\omega_2>\omega_1$, see left plot). At a much later time (bottom figure), towards the end of the experiment (which lasted $682 T_{0}$), one can see that the distribution of energy over $\ell$ and $z$ have evolved, with in particular different values of $\ell_{1}$ and $\ell_{2}$ depending on $z/H$, as well as a central region (around $z/H=0.5$), where no secondary waves are present.

Grayson et al.~\cite{Grayson:JFM:22} interpret these observations by a finite-size effect, using a model quite similar to the one presented in Bourget~et al.~\cite{Bourget:JFM:14}, taking into account how the secondary waves exit their interaction region with the primary wave due to their group velocity. However, the main difference with the model of Bourget~et al.~\cite{Bourget:JFM:14} is that the instability is allowed to be a function of space, and not only time as was the case in the previous model. The authors show that including this feature in the model actually results in the observed variability. In addition, they show that with the parameters of the experimental study, the ratio between the residence time of the secondary waves in the interaction region and the development time of the instability is close to one, so that the TRI is actually very sensitive to the actual properties of the secondary beams. As a result, preferential selection of triads is locally determined. As one triadic interaction decays, another forms, but this time at a different physical location and with new secondary beam properties. This explains why, experimentally, modulations are observed not only in physical space but also in Fourier space.



\subsection{What Happens with TRI when Combining Stratification and Rotation ?}

An extensive comparison of the characteristics of TRI between gravity and inertial waves has been presented in Section~\ref{subsection:TRI}, mainly based on the joined studies of the Lyon and Orsay groups. A question remains: how is, for example, the TRI for gravity waves modified when rotation is added? This study has been performed in~\cite{Maurer:JFM:2016}, where the same experiment used in~\cite{Bourget:JFM:13,Bourget:JFM:14} was placed on a rotating table, which was rotated at values of $f/N$ between 0 and 0.45.

\begin{figure}[t]
\includegraphics[width=0.99\linewidth]{fig5.pdf}
\caption{(Left) Experimental observations showing the triadic resonance stability or instability of a primary wave of dimensionless frequency $\omega_0/N$ at Coriolis parameter $f/N$. The instability threshold in this ($f/N,\omega_0/N$) phase space is shown by various curves, depending on the beam width. The hatched zone corresponds to a forbidden zone for TRI (see text). (Right) {Renormalized amplitude of the Fast Fourier Transform of the vertical gradient of density for various values of $f/N$. For the sake of clarity, the spectra were shifted vertically. The vertical dashed line indicates $\omega/N=0.4\simeq\tfrac{1}{2}\omega_0/N$.} On each spectrum the peak on the right represents the primary wave, daughter waves correspond to the other peaks. Extracted from~\cite{Maurer:JFM:2016}.}\label{fig:Maurer}
\end{figure}


The main results are summarised in Figure~\ref{fig:Maurer}. The left plot represents the phase space ($f/N$, $\omega_{0}/N$), where the instability threshold, computed including the effect of rotation in the finite-width analysis of Bourget~et al.~\cite{Bourget:JFM:14}, is represented by several lines, depending on the beam width (infinite, 3 wave lengths and 1.5 wave length). One can observe that - as expected since the group velocity, at the heart of the finite-width effect, decreases with the rotation rate - the instability threshold lowers when rotation increases, all the more when beams are narrow. This effect only takes place on a limited range of rotation rates, since in this phase space there is an exclusion region for TRI (hatched). This is due to the fact that, as rotation rate increases above $\omega_{0}/2$, with secondary wave frequencies that have to be larger than $f$, it implies $\omega_{0}>2f$, which is not possible for propagating waves. As a result, there is a value of $f/N$ for which the threshold is minimum, reminiscent of the critical latitude phenomenon described in MacKinnon~et al.~\cite{Mackinnon:JPO:13c}. Experimental data points are shown, with symbols indicating if TRI was observed or not, and they are in very good agreement with the analytical predictions given by the smooth curves.

On the right plot of Figure~\ref{fig:Maurer}, the frequency spectra for all values of rotation rates are shown, allowing to observe the evolution of the secondary frequencies with $f/N$. As rotation increases, the frequencies of the two secondary waves get closer (and therefore closer to $\omega_{0}/2$). As the rotation rate gets very close to the limit of the forbidden region ($\omega_{0}/2$), the secondary wave of lowest frequency surprisingly falls below $f$. An explanation of this result could be the existence of a non-zero out-of-plane component of the wave vector. This has been discussed in Mora~et al.~\cite{Mora:PRF:21}, who show that when rotation is present, the maximum growth rate is obtained with a non-zero component of the wave vectors in a direction perpendicular to the primary wave beam. This result is attributed to the 3D structure of inertial waves. Indeed, while theoretical analysis have shown that coplanar triads always display the highest growth rates for TRI of internal gravity waves~\cite{Ghaemsaidi:JFM:19}, in contrast, for inertial waves the possibility of an out-of-plane component of the secondary wave vectors is still an open question both theoretically and experimentally.



\section{Axisymmetric Waves}

Beyond two-dimensional ($2$D) wave beams, various recent studies have delved into three-dimensional ($3$D) internal waves assuming that they would exhibit peculiar dynamics. The drive to study such wave fields is, interestingly, both experimental and theoretical to investigate whether purely 3D effects may exist, either in the linear or in the non-linear regime. Besides in-situ measurements of axisymmetric wave fields~\cite{kumar2011, schubert1980}, it is interesting to note that various laboratory systems generate such wave fields, e.g. vertically oscillating spheres~\cite{stevenson1969, Flynn:JFM:03} or impacting plumes~\cite{ansong2010}.

In what follows, we will use the distinction made in~\cite{Boury:JFM:23} between axisymmetric waves that are invariant by rotation ($\theta$-independent, where $\theta$ is the orthoradial angle in the horizontal plane) and cylindrical waves that allow for an orthoradial dependence ($\theta$-dependent). Axisymmetric waves are a particular class of cylindrical waves, easier to generate and study experimentally.

Axisymmetric wave fields have traditionally been experimentally excited using a vertically oscillating sphere and exploring the shape of the wave beams~\cite{ermanyuk2011, Flynn:JFM:03, Ghaemsaidi:JFM:16, mowbray1967a, peacock2005, stevenson1969}. The radially decreasing amplitude and the viscous decay of the conical wave beam emitted by an oscillating sphere has been explored in laboratory experiments by Flynn et al.~\cite{Flynn:JFM:03} showing good agreement with theoretical predictions. More sophisticated axisymmetric experimental geometries have later been investigated using a vertically~\cite{Duran:PRE:13} and a horizontally~\cite{ermanyuk2017} oscillating torus, in which case a highly non-linear process occurs due to the three-dimensional geometric focusing, able to transport momentum and break into turbulence. None of these experimental configurations, however, readily permitted a change in the form nor the wave number of the wave field being excited. Versatile configurations have been obtained thanks to a cam-based axisymmetric wave~\cite{Maurer:EF:17} generator, derived from the planar wave generator developed by Gostiaux et al.~\cite{Gostiaux:EF:07} (see Table~\ref{tab:gene}).

The experimental study of axisymmetric waves presents several difficulties. First, the fundamentally $3$D geometry of the waves means that a complete characterisation requires visualizations both in the vertical and in the horizontal planes. A way to obtain this imaging is to elevate the tank and use an mirror inclined at $45$ degrees below it (e.g.~\cite{Maurer:EF:17}). Second, to match boundary conditions with the axisymmetry of the wave field, it is more appropriate to use a cylindrical tank with the same geometry; to prevent optical distorsions, this cylindrical tank should then be enclosed in a larger tank of square cross section.



\subsection{Linear Dynamics}

Axisymmetric waves can be described in natural cylindrical coordinates $(r,\theta,z)$ by decomposing the velocity field into two horizontal (radial and orthoradial) and one vertical velocity components, $(v_r,v_\theta,v_z)$. For practical reasons, it is instructive to introduce the stream function under a symmetric formalism
\begin{equation}\label{eq:axi1}
v_r = -\frac{1}{r}\dfrac{\partial \left(r \psi \right)}{\partial z} \hspace{1cm} \quad\text{and}\qquad v_z = \frac{1}{r}\dfrac{\partial \left(r \psi \right)}{\partial r}.
\end{equation}
Neglecting non-linear and dissipative terms, the axisymmetric stream function satisfies a collapsed equation
\begin{equation}\label{eq:axi2}
\dfrac{\partial^2 \Delta \psi}{\partial t^2} + f^2 \dfrac{\partial^2 \psi}{\partial z^2} +\frac{N^2}{r} \dfrac{ \partial^2 \left(r\psi \right)}{\partial r^2} = 0,
\end{equation}
whose solutions can be found analytically under the form of either a product of a radial Bessel function and a vertical exponential, or a product of a radial Hankel function and a vertical exponential. These are the counterparts of sine-cosine and exponentials for Cartesian modes and plane waves, respectively. Bessel-shaped wave fields describe modes; Hankel-shaped ones are more appropriate to describe converging or diverging conical wave fields.

The dispersion relation can be obtained thanks to a Hankel--Fourier transform on equation~\eqref{eq:axi2}, yielding
\begin{equation}\label{eq:axi3}
\omega^2 (\ell^2 + m^2) = f^2 m^2 + N^2 \ell^2,
\end{equation}
with $\omega$ the wave frequency, and $\ell$ and $m$ the radial and vertical wave numbers.

Cylindrical wave fields satisfy a system of equations similar to equation~\eqref{eq:axi2}, but with non-zero $\theta$-derivatives. Solutions to these equations are called Kelvin modes~\cite{Guimbard:these:08,Guimbard:JFM:10,Boury:JFM:23}. They involve rotating and counter-rotating waves, and preserve the $2\pi$ orthoradial periodicity through a dimensionless, integer, orthoradial wave number $p$. The axisymmetric dynamics is recovered by setting $p=0$. Interestingly, $p$ does not appear in the dispersion relation~\eqref{eq:axi3} which thus remains the same for both axisymmetric and cylindrical waves. A thorough description of these Kelvin modes can be found in~\cite{Boury:JFM:23}.

Horizontal axisymmetric modal wave fields, propagating vertically, are the easiest to study and characterize experimentally as the radial structure can be described under the simple analytical form of a single Bessel function. Similarly to Cartesian studies, axisymmetric modes can be obtained experimentally by enforcing appropriate boundary conditions with a tank of the same radial size as the wave generator. Such modes have been described in~\cite{Boury:PRF:19}.



\subsection{Non-Linear Dynamics}

While the linear dynamics of axisymmetric waves remains fairly similar to the linear dynamics of Cartesian plane waves, strong differences have been evidenced in the non-linear regime both in stratified and in rotating experiments~\cite{Boury:PRF:21, Boury:JFM:23, Maurer:these:17, Shmakova:JFM:19, Noir:PEPI:12}. The analytical justification is two-fold: first, contrary to the plane wave case, the cylindrical Jacobians giving the non-linear terms are no longer anti-symmetric, due to the $r$-derivatives, preventing some terms to mutually cancel out; second, while the product of two exponentials (plane waves) or sine-cosines (Cartesian modes) can be expressed under the form of another exponential or sine-cosine, the product of two Bessel functions does not usually have a similar simple expression but rather yields a sum of Bessel functions. This is further complexified for waves with a non-zero orthoradial velocity $v_\theta$ (notably inertial waves), as the horizontal Laplacian introduces additional non-linear terms coupling the horizontal velocities ($v_r$, $v_\theta$)~\cite{Boury:JFM:23}. These different terms bring several questions regarding the non-linear dynamics of axisymmetric and cylindrical waves, whether it be the subsistance of the triadic resonant instability in a similar formalism as with plane waves, or the possibility for other non-linear behaviors to arise due to the modification of the Jacobian structure.



\subsubsection{Triadic Resonant Instability}

Axisymmetric Triadic Resonant Instability (TRI) has been proposed as a natural extension of TRI, extensively studied in $2$D for plane waves. Various experimental studies in axisymmetric geometry led to the observation of localized TRI whose resonant conditions, both temporal and spatial, are the same as for plane waves~\cite{Shmakova:JFM:19, Boury:JFM:23}. This is due to the assimilation, locally, of the axisymmetric wave field with a Cartesian wave field when disregarding the overall geometry of the system imposed by the forcing or by the boundary conditions. As such, it does not provide clear indications on a totally cylindrical, fully developed, TRI that would preserve the imposed geometry of the system. In axisymmetric focusing experiments performed with an oscillating torus in a stratified fluid, \cite{Shmakova:JFM:19} have shown that a primary wave field can lead to resonant subhmarmonics in agreement with the $2$D TRI theory. However, other experiments led by Maurer et al.~\cite{Maurer:these:17} in a stratified and rotating fluid suggested that the axisymmetry could be broken for the subharmonics and that they may not fully satisfy the spatial resonance conditions.

An analytical derivation of the resonance conditions involving three cylindrical wave fields (Kelvin modes) performed by~\cite{Boury:JFM:23} showed that a fully $3$D TRI could exist, preserving the cylindrical geometry of the system, although with some modifications compared to the Cartesian case. First, the three waves have resonant frequencies $\omega$ and two exact spatial resonances, along the vertical ($m$) and orthoradial ($p$) directions. The $p$ resonance along $\theta$ is particular insofar as the wave fields also have to satisfy a $2\pi$ periodicity, which necessarily leads to a discretized resonance. Second, the situation gets more complex along the radial direction, as exact resonances have been shown to exist only for sufficiently large wave fields whereas in configurations of small extent or in confined geometries quasi-resonances are more likely to appear. Lastly, due to an asymptotic divergence at the center of the domain, it is unclear whether waves are locally spatially resonant at $r=0$.

\begin{figure}[!ht]
\includegraphics[width=1\linewidth]{fig6.pdf}
\caption{Experimental visualization of TRI for cylindrical wave fields. The top row shows the radial velocity field $v_r$ in the vertical plane; the bottom row shows the same radial velocity field in the horizontal plane. From left to right: forcing wave field, at $\omega_0 = 0.80\mathrm{~rad\cdot s^{-1}}$; first and second subharmonics at $\omega_1 = 0.50\mathrm{~rad\cdot s^{-1}}$ and $\omega_2 = 0.30\mathrm{~rad\cdot s^{-1}}$. Experiments conducted at $f=0.29\mathrm{~rad\cdot s^{-1}}$ and $N=0.97\mathrm{~rad\cdot s^{-1}}$ (adapted from~\cite{Boury:JFM:23}).}
\label{fig:axisym-TRI}
\end{figure}


These results have been confirmed experimentally both in confined and in unconfined geometries~\cite{Boury:JFM:23}. Figure~\ref{fig:axisym-TRI} shows an example of TRI obtained in an unconfined stratified-rotating experiment ($f=0.29\mathrm{~rad\cdot s^{-1}}$ and $N=0.97\mathrm{~rad\cdot s^{-1}}$), in which the forcing wave (shown in the first column) generates two resonant subharmonics (second and third columns). The velocity field in the horizontal plane shows that the two subharmonics are not axisymmetric. In addition, their time behavior shows that they have opposite orthoradial wave numbers. Interestingly, the axisymmetric forcing seems to lead to axisymmetric subharmonics if the fluid is stratified only; however, if the fluid is also rotating (as in the example shown in Figure~\ref{fig:axisym-TRI}), cylindrical subharmonics with a non-zero orthoradial wave number have been observed with the emergence of two contra-rotating subharmonic waves. Similar experiments performed in a confined domain have shown that TRI is slightly modified, as the radial wave numbers of the waves forming the triad can be quasi-resonant and not exactly resonant\cite{Boury:JFM:23}. This result is consistent with the fact that boundary conditions are known to have a significant impact on the characteristics of TRI, as discussed in the case, for instance, of surface waves~\cite{Michel2019}.



\subsubsection{Super-Harmonic Generation}

While for a single monochromatic plane wave, the self-interaction term $(\bfu\cdot\mathbf{\nabla})\bfu$ vanishes, this is no longer the case for axisymmetric wave fields. A direct consequence is that axisymmetric waves can self-interact and transfer energy to waves at higher frequencies, contrary to Cartesian plane waves (self-interaction is only possible for plane waves if the stratification is non-linear~\cite{Baker:JFM:20}). This process generates super-harmonics, i.e. waves at frequencies $2\omega$, $3\omega,\,\dots$ from an axisymmetric wave field initially forced at $\omega$. These super-harmonics are propagating if the primary forcing occurs at sufficiently small frequency so that the harmonic frequencies remain smaller than the cut-off frequency imposed either by $f$ or $N$. Evidenced for the first time in the experimental study of~\cite{Boury:PRF:21}, these harmonics remain axisymmetric if the fluid is stratified and non-rotating. The energy is then distributed onto several waves (cavity modes in the aforementioned study), that satisfy temporal resonance conditions but not necessarily spatial ones. As for TRI, this could be due to the enforcement of boundary conditions that prevail over the resonances.



\section{Internal Gravity Wave/Inertial Wave Attractor}

\subsection{\texorpdfstring{$2$}{2}D and \texorpdfstring{$3$}{3}D Reflection}

As underlined in the introduction, internal waves propagate along a double cone with a half top angle equal to $\beta$ which is given by the dispersion relation~\eqref{Disp_rel_global}. Such a property leads to a peculiar reflection law of these waves on a slope. Snell's reflection law, applicable e.g. for light waves and surface waves, prescribes that the reflected and the incident waves have the same angle with respect to the normal to the slope. This is not the case for internal waves (either gravity or inertial waves). The first conservation law in this case is given by the non penetrability condition, written as
\begin{equation}
\left(\bfv_r+\bfv_i\right)\cdot \bfn=0.
\end{equation}
This leads to the conservation of frequency and of the projection of the wave vector along the direction of the slope. As a consequence, the reflected and incident waves propagate along the same double cone.

In $2$D geometry, this means that the reflected wave propagates in a different quadrant of the Saint Andrews cross. In addition, the reflection on a slope inclined with an angle $\alpha$ with the horizontal leads to focusing or defocusing, depending on the relative values of $\alpha$ and $\beta$ and on the direction of propagation of the incident wave. When focusing occurs, the wave length of the wave beam decreases by a factor
\begin{equation}
\gamma=\left| \frac{\sin(\alpha+\beta)}{\sin(\alpha-\beta)}\right|\,.
\end{equation}
Assuming conservation of energy, the energy flux density is increased by a factor $\gamma^2$ and the reflected wave is therefore more prone to instability. Note that the case $\beta=\alpha$ is a singularity ($\gamma\rightarrow \infty$), the so-called critical reflection. It has been the focus of theoretical analysis~\cite{Dauxois:JFM:99} and experiments~\cite{Horne:EF:21}.

In three dimensions, the orthoradial angle is not imposed by the dispersion relation and is therefore not conserved. The derivation of the reflection law in this case has been performed for internal gravity and inertial waves following slightly different approaches~\cite{Phillips1966,Eriksen1982,MandersMaas:2004, Pillet:EJM:2019,Favier:JFM:2024}. The different derivations use three ingredients: non penetrability; propagation of the incident and reflected waves on the same cone and conservation of the horizontal component of the velocity normal to the slope. They have shown that the rays of waves change direction upon reflection and have the tendency to converge toward the upslope gradient.



\subsection{Wave Attractor}

In a closed domain with a particular geometry and for a specific range of frequencies, successive reflections of wave rays in the domain have been shown to converge on the same limit cycle, which is a geometric pattern, called wave attractor~\cite{Maas:JFM:95}. Theoretical predictions can be made using ray-tracing methods, in which the wave beam is assumed to be localized enough so that its propagation along the group velocity can be described by a geometrical ray path. The phenomenon of concentration of the energy on attractors in $2$D geometries has been the subject of different mathematical studies~\cite{MandersMaas:2003,BajarsMaas:JFM:2013,deVerdiere-StRaymond:2020,MakridinSibgatullin:PRF:2023}. In the past years, different geometries leading to the possible existence of wave attractors have been investigated. These works have been summarized in a recent review~\cite{Sibgatullin:2019aa}.

In astrophysics, the link between wave attractors and trapped wave in the equatorial regions of rotating planets has been first considered~\cite{Stern1963,Bretherton1964, Stewartson1972}. Rotating spherical shells have been shown to be a generic geometry for inertial wave attractors~\cite{RieutordValdettaro1997,RieutordetalPRL2000}, which appear in numerical simulations~\cite{Ogilvie:Astro:2004}. The role of different features such as the behavior of the waves in the vicinity of the axis of rotation, the critical latitude phenomenon, and non-linearities have then been analyzed~\cite{RieutordValdettaro2010, Rabitti:JFM:2013, Favier2014, He:JFM:2023}.

Even if different works have considered the possible existence of internal wave attractors in real stratifications and topographies using different geometries~\cite{MandersMaas:JGR:2004,Hazewinkel:JFM:10,TangPeacock:Chaos:2010,Wang:Oceanologica:2015}, no field observations of attractors has been reported so far.

In the following, we will focus on the main experimental results obtained in 2D trapezoidal geometries and in an axisymmetric frustrum to show the difference and similarities between gravity and inertial wave attractors.



\subsection{Experiments in Quasi-\texorpdfstring{$2$}{2}D Trapezoidal Domains}

Following his seminal theoretical study~\cite{Maas:JFM:95}, Maas has collaborated in different experimental studies to visualize internal gravity wave and inertial wave attractors. There were first observed in a density stratified fluid using an oscillating tank~\cite{Maasetal1997}. Even though the structure of an inertial wave is three-dimensional, a linear attractor was also observed in the case of inertial waves, using as well an oscillating tank as generator~\cite{maas2001, MandersMaas:2003, MandersMaas:2004}. Without dissipation, the energy density along the attractor tends to infinity due to multiple focusings. In a more realistic scenario, the energy is dissipated through viscous dissipation or non-linear phenomena. The balance between focusing at the slope and viscous effects can be used to predict the width of the branches of the attractor. Based on experimental results for gravity internal waves~\cite{HBDM2008}, this width was predicted to scale as $(\nu/L^2N)^{1/3}$ with $L$ the perimeter of the attractor. This result was checked numerically by Grisouard et al.~\cite{Grisouardetal2008}. For inertial waves, the equivalent scaling factor is $\rEk^{1/3}=(\nu/L^2 f)^{1/3}$, as verified numerically~\cite{JouveOgilvie2014, Wu:JFM:2023}.

To illustrate this discussion, a typical linear wave field for both types of attractors is displayed in Figure~\ref{fig:2Dattractor}(left). The scaling proposed above was satisfied in the case of a linear inertial wave attractor~\cite{Brunet:PRF:19}. For the linear gravity wave attractor, the agreement was less clear~\cite{Brouzet:JFM:16}. An explanation was further proposed in~\cite{Beckebanze_Brouzet_Sibgatullin_Maas_2018} and validated in~\cite{Davis:JFM:19}. It is related to boundary dissipation effects, whose magnitude is different between the two configurations, due to the size of the tanks. The different tanks were chosen to be adapted to the structure of the waves. For internal gravity waves, the tank had a small thickness compared to the other dimensions and the velocity profiles were measured to be two-dimensional except in the viscous boundary layer near the lateral walls. This layer induces a strong dissipation which is at least ten times larger than the bulk dissipation. This effect is reduced in the case of inertial waves, since the tank is much wider, relatively to the other typical dimensions, and as a consequence, the bulk dissipation dominates. A new scaling law for the size of the branches of the internal wave attractors was then proposed and is in better agreement with experiments~\cite{Beckebanze_Brouzet_Sibgatullin_Maas_2018}.

\begin{figure}[t]
\includegraphics[width=0.9\linewidth]{fig7.pdf}
\caption{Examples of experimental realization of internal wave attractors. (Top left) Synthetic Schlieren visualization of a linear gravity wave attractor (adapted from~\cite{Brouzet:JFM:16}). (Top right) Kinetic energy for a non-linear gravity wave attractor (adapted from~\cite{Brouzet:EPL:16}). (Bottom left) Horizontal velocity field for a linear inertial wave attractor (adapted from~\cite{Brunet:PRF:19}). (Bottom right) Horizontal velocity field for a non-linear inertial wave attractor (adapted from~\cite{Brunet:PRF:19}).}
\label{fig:2Dattractor}
\end{figure}

\break
The three-dimensional effects on a wave attractor were then studied for internal gravity waves by Pillet et al.~\cite{Pilletetal2018} using an extension in the third direction of the quasi-$2$D experiments performed at ENS de Lyon. These experiments show a trapping of the waves in different $2$D planes.

Due to the accumulation of energy in the branches of the attractor, gravity wave attractors and inertial wave attractors are prone to instabilities and in particular triadic resonance instability. Snapshots of the non-linear gravity wave attractor obtained at ENS de Lyon~\cite{Brouzet:JFM:16, Brouzet:EPL:16, Brouzet:JFM:17} and of the non-linear inertial wave attractor obtained at Paris-Saclay~\cite{Brunet:PRF:19} are reproduced in Figure~\ref{fig:2Dattractor}. One can still see the geometrical structure, on top of which several other waves have formed. Both studies have shown that the attractor amplitude was reduced and the attractor typical length increased.

Such results were attributed to an energy transfer due to the instability and can be quantified by replacing, in the model for linear attractors, the kinematic viscosity by a turbulent viscosity based on the features of the forced waves~\cite{Brunet:PRF:19}. In the non-linear case, strong differences between gravity wave attractors and inertial wave attractors have been observed. An important one is the selectivity of the secondary waves. The feature already mentioned for wave beams (Section~\ref{subsection:TRI} and Figure~\ref{fig:bordesbourget}) is also observed here: indeed, for inertial waves the frequency peaks for the secondary waves are very wide compared to the ones for gravity waves. In addition, there is a clear discrepancy between experimental results for rotating fluids and the 2D numerical simulations~\cite{JouveOgilvie2014}, while the agreement between experiments and simulations is very good for gravity wave attractors~\cite{Brouzet:JFM:16}. This weak selectivity in the case of inertial waves can be attributed to the three-dimensional structure of these waves interacting with physical walls, which was clearly neither the case for internal gravity waves~\cite{Brouzet:JFM:16}, nor in the numerical simulations of inertial waves using proper boundary conditions~\cite{JouveOgilvie2014}. The second difference concerns the possibility of secondary inertial waves with a non-zero component in the direction perpendicular to the plane of the attractor. This observation, specific to inertial waves and already mentioned for wave beams, could explain the dependence of the secondary waves on Reynolds number, which displays an opposite behavior compared to 2D numerical simulations. Indeed, in the latter case the frequencies $\omega_1$ and $\omega_2$ of the subharmonic waves tend toward $\omega_0/2$ for larger Reynolds number while for experiments they depart from $\omega_0/2$.



\subsection{Axisymmetric Domain: Frustrum}\label{sec:frustrum}

A second geometry has been widely used to study 3D effects for gravity or inertial wave attractors: the frustrum, i.e. an annulus whose inner cylinder is replaced by a truncated cone. This geometry has been investigated for inertial waves both numerically and experimentally~\cite{Swart:DAO:10,klein2014,Sigbatullin:AIP:2019,Boury:JFM:21}. These studies show a wide variety of dynamical systems including linear wave attractors, triadic instability, and emergence of mean flows.

We would like to focus on a similar geometry without the inner cylinder, used in experiments conducted at ENS de Lyon~\cite{Pacary:PRF:23}. Using exactly the same experimental setup, rotating fluids and stratified fluids were studied. If the formation of a linear internal wave axisymmetric attractor is similar for both inertial and gravity waves, a major difference is visible for the nonlinear dynamics. Here again, the secondary waves are associated with discrete frequencies for gravity waves contrary to inertial waves for which the frequency range is large, as can be seen in Figure~\ref{fig:pacary} (top, respectively left and right). To explain this behavior, the authors of this study~\cite{Pacary:PRF:23} show that the secondary internal waves are the global resonant modes of the tank while the secondary inertial waves are only localized near the center of the tank, as can be seen in the bottom row of Figure~\ref{fig:pacary}, left and right respectively. It is explained by the fact that for rotating fluids, standing modes are associated with frequencies larger than the forcing frequency and therefore not accessible through TRI, resulting only in a local appearance of these secondary waves. The important role of the boundaries is therefore highlighted in the respective non-linear dynamics of gravity and inertial waves. At Paris-Saclay, an experiment to study internal wave turbulence has been designed with inclined panels at the bottom of the tank to inhibit the emergence of wave eigenmodes~\cite{Lanchon:PRF:2023}. In such a configuration, the frequency peaks of the secondary waves excited through TRI are much broader. The frequency spectrum for internal gravity waves becomes therefore closer to the frequency spectrum observed for weakly non-linear inertial waves (Figure~\ref{fig:bordesbourget} right).

\begin{figure}[ht]
\includegraphics[width=0.9\linewidth]{fig8.pdf}
\caption{Non-linear behavior of an axisymmetric wave attractor in a frustrum, for gravity waves (left) and inertial waves (right). (Top row) Frequency spectrum. (Bottom row) Snapshots of one of the secondary waves filtered at a given frequency. (Figure adapted from~\cite{Pacary:PRF:23}).}\label{fig:pacary}
\end{figure}

Finally, breaking the axisymmetry could induce the trapping of a wave attractor in a particular plane in 3D. This has been recently predicted theoretically and verified numerically~\cite{Favier:JFM:2024}.

%\break

\section{Conclusions and Future Directions}

In this article, we have discussed several examples of experiments investigating both the linear and non-linear behaviors of internal gravity waves in stratified fluids and of inertial waves in rotating fluids. More specifically, we have focused on the Triadic Resonant Instability (TRI) in various configurations. All these experiments show that, although these waves have a similar dispersion relation, their spatial structure is very important and must be taken into account in order to accurately predict their linear and non-linear dynamics. In particular, clear differences have been noted in the weakly non-linear regime for different geometries. The main difference is the attraction of energy by discrete frequencies associated to resonances and standing modes in the stratified case. This process seems not to be at play in the rotating case, probably because of the $3$D structure of wave field. This observation is of great interest to guide future developments in the study of internal waves.

Recently, efforts have been made to observe, in laboratory experiments, both a weakly non-linear regime of internal gravity wave turbulence in a density stratified fluid~\cite{Cortet:CRAS:2024}, and a regime of inertial wave turbulence in a rotating flow~\cite{GaltierBook}. It will be interesting as well to study the similarities and differences in these wave turbulence regimes between gravity and inertial waves.

As mentioned in the introduction, rotating-stratified fluids are ubiquitous in geophysical and astrophysical systems. Laboratory experimental studies combined with theoretical developments and numerical simulations help complement the fundamental understanding of the non-linear behavior of both internal gravity and inertial waves, and therefore help improve the parametrization of (stratified) mixing and other small scale phenomena in climate models. To further assess the role of these waves in atmospheric and oceanic processes, additional features have recently been the focus of a more sustained attention such as the generation of inertial or internal gravity waves due to moving obstacles, the combination of rotation and stratification, the role of a non-linear stratification in density stratified fluids, ice-melting and double diffusive mechanisms, wave-turbulence interaction, wave-mean flow interaction, and the coupling between internal gravity/inertial waves with other type of waves (Lee waves, Kelvin waves, surface waves\dots). These are important processes that need to be further understood.



\section*{Declaration of interests}
The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\printbibliography
\end{document}
